Author's response to reviews

Title: Accrual and drop out in a primary prevention randomised controlled trial: qualitative study.

Authors:

Helen C Eborall (hce3@le.ac.uk)
Marlene CW Stewart (marlene.stewart@ed.ac.uk)
Sarah Cunningham-Burley (sarah.c.burley@ed.ac.uk)
Jackie F Price (J.Price@ed.ac.uk)
F Gerry R Fowkes (Gerry.Fowkes@ed.ac.uk)

Version: 3 Date: 29 September 2010

Author's response to reviews:

Dear Editors

We thank you for your response to the revised version of our manuscript. We have changed the sentence identified by reviewer 2 as problematic, and hope that it is now less confusing.

As before, we have attached two versions of the revised manuscript – one with the tracked change shown and one with the change accepted. Please contact me if there is a problem accessing both versions.

We look forward to hearing from you.

With kind regards,

Helen Eborall